Author’s response to reviews

Title: Coix lacryma-jobi var. ma-yuen Stapf sprout extract has anti-metastatic activity in colon cancer cells in vitro

Authors:

Eun Suk Son (eunsuk0607@hanmail.net)
Young Ock Kim (kyo9128@korea.kr)
Chun Geon Park (pcg@korea.kr)
Kyung Hun Park (blueour@korea.kr)
Sung Hwan Jeong (jsw@gilhospital.com)
Jeong Woong Park (jwpark@gilhospital.com)
Se-Hee Kim (sehee0423@gilhospital.com)

Version: 3 Date: 18 Oct 2017

Author’s response to reviews:

Comment: Please refer to all authors by their initials in the Authors’ contribution section. Additionally, please confirm in this section that all authors have read and approved of the final manuscript.

Response: We referred to all authors by their initials in the Authors’ contribution section and confirmed in this section that all authors have read and approved the final manuscript as follows:

Authors’ contributions

E.S.S. and Y.O.K. equally contributed to the research; C.G.P. and K.H.P. provided Coix lacryma-jobi var. ma-yuen Stapf sprout extract for this research; S.H.J. analyzed the data; J.W.P. and S.-H.K. wrote the manuscript and supervised this work. All authors read and approved the final manuscript.